Author's response to reviews

**Title:** Evaluation of computer-tailored health education (‘E-health4Uth’) combined with personal counselling (‘E-health4Uth + counselling’) on adolescents’ behaviours and mental health status: design of a three-armed cluster randomised controlled trial

**Authors:**

Rienke Bannink (r.bannink@erasusmc.nl)
Evelien Joosten - van Zwanenburg (ejoosten@ggdzhz.nl)
Petra van de Looij - Jansen (pmvandelooijjansen@rotterdam.nl)
Els van As (e.van.as@rivas.nl)
Hein Raat (h.raat@erasusmc.nl)

**Version:** 5 **Date:** 10 December 2012

**Author’s response to reviews:** see over
Concerning changes made of the manuscript "Evaluation of computer-tailored health education ('E-health4Uth') combined with personal counselling ('E-health4Uth + counselling') on adolescents’ behaviours and mental health status: design of a three-armed cluster randomized controlled trial"

Dear J. Dizzon,

The next change is made to the formatting of the manuscript:
- The additional files 1-4 (ethics and funding) are deleted.

Thank you for your consideration of our work. We are looking forward to your response.

Sincerely yours on behalf of all authors,

R. Bannink, MSc.

Corresponding author:
Rienke Bannink
Erasmus MC – University Medical Centre Rotterdam
Department of Public Health
PO box 2040
3000 CA Rotterdam
The Netherlands
Email: r.bannink@erasusmc.nl
Phone: +31 10 70 44634